

Reviewer's comments:

#1 What does it mean "The tumor was removed" in the section of Laboratory examinations and imaging examinations. Did the patient receive the operation of colon resection?

**Answer:** "The tumor was removed" was revised as "Part of the tumor was removed for biopsy." The patient did not receive the operation of colon resection.

#2 There are no mention about Fig.1F in the text of Laboratory examinations and imaging examinations as well as Figure legend.

**Answer:** Fig.1F was unnecessary, we had deleted it.

5 Issues raised:

(1) The "Author Contributions" section is missing. Please provide the author contributions;

**Answer:**The "Author Contributions" section was between "**CONCLUSION**" and "**References**".

#### **Authorship Contributions**

Lei Wang and Dali Cai guided the diagnosis and treatment of this case. Lei Wang and Na Lin reviewed all related literature and analyzed the clinical data. Lei Wang drafted the manuscript. Na Lin critically revised the manuscript. All authors read and approved the final manuscript.

(2) The authors did not provide the approved grant application form(s). Please upload the approved grant application form(s) or funding agency copy of any approval document(s);

**Answer:**The approved grant application form was uploaded.

(3) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

**Answer:** Original pictures using powerpoint were uploaded.